Author's response to reviews

Title: Smoking habits in relation to the clinical picture of women with microscopic colitis compared to controls

Authors:

Bodil Roth (bodil.roth@med.lu.se)
Rita J Gustafsson (rita.gustafsson@med.lu.se)
Bengt Jeppsson (bengt.jeppsson@med.lu.se)
Jonas Manjer (jonas.manjer@med.lu.se)
Bodil Ohlsson (bodil.ohlsson@med.lu.se)

Version: 2 Date: 5 May 2013

Author's response to reviews: see over
Dear Editor,

A competing interests section have been included in the manuscripts entitled “Smoking habits in relation to the clinical picture of women with microscopic colitis compared to controls”

I would be happy to have the manuscript transferred to BMC Women's Health for further consideration.

Sincerely
Rita J Gustafsson